Author's response to reviews

Title: Moderate mental illness as a predictor of chronic disease prevention and screening

Authors:

Ginetta Salvalaggio (ginetta@ualberta.ca)
Christopher Meaney (christopher.meaney@utoronto.ca)
Rahim Moineddin (rahim.moineddin@utoronto.ca)
Eva Grunfeld (eva.grunfeld@utoronto.ca)
Donna Manca (dpmanca@ualberta.ca)

Version: 2 Date: 29 May 2017

Author’s response to reviews:

Thank you for reviewing our revised manuscript and for the opportunity to respond. I and my co-authors have addressed the suggestions for revision as follows:

1. Trial registration number (for original trial on which dataset this secondary analysis relies) has been moved from abstract to methods section, page 8 line 146.

2. Authorship declaration now includes confirmation that all authors have read and approve of the final manuscript, page 15 line 286.

3. A clean manuscript version without annotations has been uploaded.

We hereby submit our revisions for your consideration and look forward to your response.